Math 481/546 - Advanced Calculus II Spring, 1999 B. Bukiet ffl Text: Calculus – Spivak ffl Text: Advanced Calculus – Widder ffl Times: Class meets Mondays and Wednesdays 10:00-11:25 AM ffl Office Hours: Monday, Wednesday 9:00-10:00 AM and 11:30 AM - 12 Noon ffl Office: 518 Cullimore Hall ffl Phone: (973)-596-8392 ffl e-mail: bukiet@model.njit.edu ffl Grading: There will be 2 in-class examinations occurring approximately during the 6th and 11th weeks of the semester. Together, they will account for 55% of your grade. There will be no make-up exams without prior approval or doctor’s note. The final exam will account for 30% of your grade. Fifteen percent of your grade will be based on class participation, homework and projects.

Course Outline

ffl Infinite Sequences (Spivak Chapter 22) ffl Infinite Series (Spivak Chapter 23) ffl Uniform Convergence and Power Series (Spivak Chapter 24) ffl Partial Differentiation (Widder Chapter 1) ffl Applications of Partial Differentiation (Widder Chapter 4) ffl Multiple Integrals (Widder Chapter 6) ffl If time allows: Line and Surface Integrals (Widder Chapter 7)